FUNCTIONAL EFFECTS OF BOTULINUM TOXIN TYPE A IN THE HIP ADDUCTORS AND SUBSEQUENT STRETCHING IN PATIENTS WITH HEREDITARY SPASTIC PARAPLEGIA

Bas J. H. VAN LITH, MSc¹, Jasper DEN BOER, PT, PhD¹, Bart P. C. VAN DE WARRENBURG, MD, PhD², Vivian WEERDESTEYN, PT, PhD¹ and Alexander C. GEURTS, MD, PhD¹

From the ¹Department of Rehabilitation and ²Department of Neurology, Donders Institute for Brain, Cognition and Behaviour, Radboud University Medical Center, Nijmegen, The Netherlands

Objective: To investigate the functional effects of bilateral botulinum toxin A treatment and subsequent stretching of spastic hip adductors on gait and reactive lateral stepping responses in patients with pure hereditary spastic paraplegia.

Design: Explorative pre-post intervention study.

Patients: Twenty-five patients with pure hereditary spastic paraplegia.

Methods: Patients were treated with bilateral botulinum toxin A injections in the hip adductors and performed daily self-administered stretching exercises for 16 weeks. Before the intervention (T0), and 6 (T1) and 16 (T2) weeks thereafter, gait width, gait speed, and leg angles at first stepping-foot contact after lateral balance perturbations were assessed, as well as the corresponding success rates of reactive lateral steps.

Results: Compared with baseline, gait width increased by 12.6% and 9.7% and comfortable gait speed by 8.3% and 11.5% at T1 and T2, respectively. In known perturbation directions, leg angles increased by 5.9% at T1 and 8.0% at T2, while success rates increased from 70% at baseline to 90% at T1 and T2. No effects were found for maximal gait speed or lateral stepping responses in unknown perturbation directions.

Conclusion: Bilateral botulinum toxin A treatment and subsequent stretching of the hip adductors may improve gait and reactive lateral stepping in patients with pure hereditary spastic paraplegia.

Key words: hereditary spastic paraplegia; botulinum toxin; balance control; gait.

Accepted Mar 29, 2019; Epub ahead of print Apr 10, 2019

J Rehabil Med 2019; 51: 434-441

Correspondence address: Bas van Lith, Department of Rehabilitation, Donders Institute for Brain, Cognition and Behaviour, Radboud University Medical Center, PO Box 9101, 6500 HB Nijmegen, The Netherlands. E-mail: Bas.vanLith@Radboudumc.nl

Patients with pure hereditary spastic paraplegia (HSP) show a slow, retrograde axonal degeneration of the corticospinal tract, which leads to bilateral progressive lower extremity spasticity. Although muscle strength and somatosensation are often also affected, the degree of paresis and sensory loss is usually milder than in other conditions with spastic paraparesis (e.g.

LAY ABSTRACT

Patients with pure hereditary spastic paraplegia often have hip adductor spasticity, causing reduced gait width and a narrow base of support. In addition, they experience difficulties stepping sideways to recover from sideward balance perturbations. A small base of support and the presumed side-stepping difficulties probably increase their risk of falling. Botulinum toxin type A injections are commonly used to reduce spasticity in patients with hereditary spastic paraplegia, but no studies have systematically investigated the effects of botulinum toxin type A in (with subsequent stretching of) the hip adductors on gait and balance. The results of this study indicate that, after botulinum toxin type A administration, there is an increase in comfortable gait speed and gait width. Furthermore, patients show improved reactive stepping responses when the perturbation direction is known. It is concluded that botulinum toxin type A treatment and subsequent stretching of the hip adductors may improve gait and reactive lateral stepping in patients with pure hereditary spastic paraplegia.

spinal cord injury or multiple sclerosis) (1, 2). Hip adductor spasticity is often prominently present in patients with pure HSP. Besides loss of gait propulsion and deterioration of balance in the plane of progression (1, 3), patients are usually also affected by frontal-plane imbalance, to which hip adductor spasticity is suggested to be an important contributing factor (4). This may be explained by spasticity of the hip adductors leading to forced, spontaneous narrowing of the base of support while standing and walking ("scissoring"). In addition, patients with HSP may experience difficulties stepping sideways to recover from lateral balance perturbations, as the hip abductors have to overcome the involuntary activity of the spastic hip adductors. The small base of support in combination with the presumed difficulties in side-stepping probably increase the risk of falling in patients with HSP.

Botulinum toxin type A (BTX-A) injections are commonly used for reducing spasticity in patients with HSP (2, 5–8), although BTX-A is still on off-label prescription in HSP for all commercially available toxins. BTX-A treatment of the hip adductors, calf muscles and hamstrings has been suggested to lead to functional improvements in balance and gait capacities Journal of Rehabilitation Medicine

(4, 5). However, despite the high prevalence of hip adductor spasticity in patients with HSP, there are no studies that have systematically investigated the effects of BTX-A injections in these muscles, either on clinical indicators of spasticity, or with respect to balance and gait capacity. In a few previous studies with limited numbers of patients with HSP, the hip adductors were injected in some of the participants (n=5-12), yet often in combination with other muscle groups (i.e. calves, tibialis posterior, iliopsoas and rectus femoris) (6, 8, 9). Furthermore, these studies did not include stretching exercises following BTX-A treatment, as recommended by international consensus (10). Hence, the specific effects of BTX-A injections in the hip adductors with subsequent stretching exercises remain to be established.

This exploratory study focused on the effects of BTX-A injections in spastic hip adductors and subsequent stretching exercises in patients with pure HSP, using 2 primary outcomes: (i) gait width; and (ii) the quality of sideways reactive stepping responses following lateral balance perturbations. In addition, comfortable and fast gait speed, success rates of the lateral stepping responses, and various clinical (physical and functional) tests served as secondary outcome measures. The physiological effect of BTX-A usually reaches its maximum 6 weeks after the injections (11) and diminishes progressively until approximately 16 weeks after the injections. Therefore, we hypothesized that reduced hip adductor tone would translate into improvements in both gait width and lateral stepping at 6 weeks post-treatment, whereas a reduction in these effects was expected at 16 weeks after treatment.

METHODS

adductor spasticity known at the expert centre for genetic movement disorders of the Donders Institute for Brain, Cognition and Behaviour, Radboud University Medical Center, Nijmegen, The Netherlands. In addition, active recruitment took place through the national patient organization "Spierziekten Nederland". Inclusion criteria were: (i) a "pure" form of autosomal dominant HSP (either genetically proven, or based on family history); (ii) 18 years of age or older; (iii) bilateral hip adductor spasticity (Modified Ashworth Scale (MAS) score 1-4); (iv) balance- and/or gait-related activity limitations in daily life; (v)able to walk > 50 m independently with (adapted) shoes and/or orthoses (but without walking aids, such as crutches or a walker); and (vi) comfortable gait velocity >0.4 m/s. Participants were excluded if they had any cognitive impairment, or they had any comorbidity affecting their gait capacity. In addition, the final BTX-A treatment of the hip adductors should have been administered longer than 6 months before the first mea-

surement. Regular BTX-A treatments of other muscle groups

should have been performed either within 3-4 weeks or longer

than 4 months before the first measurement in order to have

Participants Participants were recruited from all patients with HSP and hip either an optimal or, otherwise, an absent effect of these prior injections during the study period.

For participants recruited at the outpatient departments, all inclusion criteria were checked by the attending physician. Participants who responded to the recruitment letters via the patient organization were interviewed by telephone by the primary investigator to check inclusion criteria i, ii and iv, whereas inclusion criteria ii, v and vi were checked by a physiotherapist at the first visit for the study. Out of 75 patients, 25 fulfilled the inclusion criteria and were included. Their demographic and clinical characteristics are listed in Table I.

Ethics statement

All subjects gave their written informed consent prior to participation. The study was approved by the regional medical-ethics committee and conducted in accordance with the Declaration of Helsinki. Due to lack of prior research on gait width and quality of reactive sidesteps in patients with HSP, no formal power calculation could be performed. Given the exploratory nature of the current study, a required number of 25 patients seemed optimal, feasible and justified, and was agreed upon by the medical ethics committee.

Intervention

Each participant was treated with bilateral BTX-A injections in the hip adductors by 1 of 3 rehabilitation physicians of our university hospital. A solution of 100 U Xeomin® (Botulinum toxin type A, Merz Pharmaceuticals GmbH, Frankfurt am Main, Germany) per 5 ml saline 0.9% was used. A dose of 150 or 200 U per leg was injected, depending on the degree of hypertonia (Modified Ashworth Scale (MAS) 1: 150 U; MAS \geq 2: 200 U).

Table I. Patient	characteristics	before the	intervantion	(T0)
	cilaracteristics		inter vanteion	

Characteristics	
Sex, male/female, n	12/13
Gene, n	
SPG-4	12
SPG-10	2
SPG-17	1
SPG-31	2
AD, genotype not confirmed	8
Tibialis anterior, median (range)	
MAS	0 (0-1)
MRC	5 (2-5)
Triceps surae, median (range)	
MAS	
knee extended	2.5 (1-4)
knee flexed	2.5 (0-4)
MRC	5 (2-5)
Hip adductors, median (range)	
MAS	3 (1-4)
MRC	5 (2-5)
Hip abductors, median (range) MRC	5 (2–5)
BBS, median (range)	49.5 (27-56)
Age, years, mean (range)	53.5 (26-72)
Hip abduction, mean (range)	
ROM	38.8 (20.0-52.5)
6MWT, mean (range)	367.5 (196-515)
TUG, mean (range)	11.1 (6.3-23.0)
ABC, mean (range)	49.83 (17-95.3)

AD: autosomal dominant inheritance; MAS: Modified Ashworth Scale; MRC: Medical Research Council scale; BBS: Berg Balance Scale; ROM: range of motion; 6MWT: 6-min Walk Test; TUG: Timed Up and Go test; ABC: Activities-specific Balance Confidence scale.

Table II. Distribution of botulinum toxin per leg

	Percentage thigh	Units/ml	
	length*, %	MAS 1	$MAS \geq \! 2$
Gracilis	15	15/0.75	20/1.00
	35	15/0.75	20/1.00
	60	15/0.75	20/1.00
Adductor magnus	15	30/1.50	40/2.00
	35	30/0.75	40/2.00
Adductor longus	15	30/1.50	40/2.00
	35	15/0.75	20/1.00

*Upper leg length was calculated from pubis bone to medial femoral condyle. Thigh-length percentages were expressed from pubis.

The BTX-A was distributed over the gracilis, adductor magnus and adductor longus muscles according to Table II. Muscle selection was based on muscle volume and its moment arm with respect to the hip joint. Thigh length was measured from the pubic bone to the medial femoral condyle and thigh length percentages were calculated from the pubic bone (see Fig. 1). All BTX-A injections were placed under ultrasound guidance.

During the 16-week study period, participants were instructed to perform stretching exercises of the hip adductors (with the hips both flexed and extended) for approximately 10 min, 3 times per day, and to log their exercises in a diary. The exercises were individually demonstrated and instructed by a physiotherapist at the day of the BTX-A injections until each participant was able to correctly perform the exercises independently.

Outcome measures

Instrumented gait assessments. For instrumented gait assessment, the GAITRite system was used (CIR Systems, Inc., Sparta, NJ, USA), which is a 4.88-m long carpeted walkway that contains pressure sensors that detect the position of each footfall (12). Participants started at one end of the GAITRite and were instructed to walk 3 times across the walkway with (adapted) shoes and/or orthoses at their preferred speed. Subsequently, they walked 3 times across the walkway at their maximal speed without risking a fall. For each step, gait width was determined by the GAITRite system (and stored as "stride width") and exported to Microsoft Excel. For each participant, the median gait width of all steps was calculated to avoid a



Fig. 1. Schematic view of injection locations in percentages thigh length from the pubic bone to the medial femoral condyle. At the indicated thigh lengths, botulinum toxin (BTX) was injected into the adductor longus (L), adductor magnus (M) and gracilis (G).

disproportional influence of a single outlying step. Gait width was determined for both the preferred (primary outcome) and maximal gait speed. In addition, the mean preferred and mean maximal gait speed of the 3 trials was calculated.

Instrumented dynamic balance assessments. For instrumented dynamic balance assessments, the Radboud Falls Simulator was used (RFS). The RFS is a moveable platform $(240 \times 174 \text{ cm}; \text{BAAT}, \text{Enschede}, \text{The Netherlands (13)})$ that can translate in multiple directions. In this way, perturbations can be imposed that mimic natural situations. In this study, only sideways perturbations were used, where a leftward platform translation resulted in a rightward balance perturbation and vice versa. In the remainder of this text, we consistently refer to the direction of the balance perturbations.

At the start of each measurement, participants were instructed to sustain all perturbations by making a single lateral step without grabbing the handrails, while the perturbation direction was known to the participants. All participants wore a safety harness attached to the ceiling that prevented them from falling. In addition, a railing system was present at both sides that participants could grab in the case of a fall. At the first measurement (T0), participants were exposed to increasing perturbation intensities that started at 0.125 m/s² and were increased by at least 0.125 m/s² between trials. The maximum perturbation intensity that each participant could successfully sustain with a single step at least once out of 3 trials (without falling or grabbing the railing system) defined the individual limit of stability. To familiarize them with the test situation, during the subsequent measurements (T1 and T2), participants were again exposed to at least 15 incremental perturbation intensities until their individual limits of stability were reached.

During all measurements each participant was twice (randomly) exposed to 5 leftward and 5 rightward perturbations at their individual limits of stability. During the first 10 perturbations, the perturbation direction was known to the participant, whereas during the last 10 trials the perturbation direction was unknown. These 20 trials were used for statistical analysis.

During all balance assessments, kinematic data were recorded by an 8-camera 3D motion analysis system (Vicon Motion Systems, Oxford, UK) at a sample rate of 100 Hz. Reflective markers were placed at anatomical landmarks according to the full-body PlugInGait configuration (14). For all trials in which the participant succeeded to make a sidestep following the perturbation, the leg angle (primary outcome) was calculated at the instant of stepping-foot contact, as the body configuration at step contact appears to critically determine the successfulness of balance recovery responses (15). The leg angle was defined as the angle between the absolute vertical and a line connecting the mid-pelvis and the ankle marker of the stepping foot. The medians of the 5 right and 5 left leg angles were calculated to avoid a disproportional influence of a single outlying value. These medians were averaged for each person into a single leg angle score for the perturbations with known and unknown directions separately. Furthermore, the success rates of the lateral stepping responses were calculated for the perturbations with known and unknown directions separately. A trial was scored as successful if the participant maintained balance with a single sidestep.

Physical tests. Muscle strength of the hip adductors and hip abductors was assessed with the Medical Research Council (MRC) scale (0–5), with lower scores indicating more muscle weakness (16). Muscle tone of the hip adductors was assessed using the MAS (0–5), with higher scores indicating more hypertonia (17). Furthermore, passive range of motion (ROM) on

hip abduction was measured using a goniometer. All outcomes were averaged for both sides into a single score.

Functional tests. Functional balance was assessed barefooted with the Berg Balance Scale (BBS, range 0-56) (18). The Timed Up and Go (TUG test) (19) was performed with (adapted) shoes and/or orthoses, but without other walking aids. Endurance was assessed with the 6-min Walk Test (6MWT) (20) during which all walking aids were allowed. Lastly, the Activities-specific Balance Confidence (ABC) scale (21) was obtained as a subjective measure of mobility and balance.

Procedure

All outcome measurements were performed on the day of treatment prior to the injections (baseline; T0), $6 (\pm 1)$ weeks after treatment (T1), and 16 (± 1) weeks after treatment (T2). T1 was set at 6 weeks after treatment, because at this time-point the physiological effects of BTX-A on spasticity were expected to have reached a maximum (11). The effects of BTX-A were expected to diminish progressively until approximately 16 weeks after the injections. Hence, T2 was set at 16 weeks posttreatment to test the possible presence of a long-term effect of the combined treatment. The instrumented assessments were taken by the primary investigator. All other tests were assessed by an independent physiotherapist.

Statistical analysis

All instrumented (parametric) outcome measures were tested using repeated-measures analysis of variance (ANOVA). The primary outcomes gait width and sidestep leg angle were tested using time (T0 - T1 - T2) as within-subjects factor. For gait width and gait speed, also gait condition (preferred - maximal) was introduced as a within-subjects factor. Bonferroni corrected post-hoc tests were applied in the case of a significant main effect of time. All other (non-parametric) outcome measures (physical

Table III. Outcomes measures

and functional tests) were analysed with a Friedman's test using time (T0 - T1 - T2) as a within-subject factor. Wilcoxon's tests were used for post-hoc comparisons in the case of a significant time effect. All statistical analyses were performed using IMB SPSS Statistics Version 22 for Windows. The α -level was set at 0.05 for all analyses, with no adjustment for multiple outcomes.

RESULTS

Participants

From the total of 25 patients included in the study, 3 patients were lost between the first (T0) and second measurements (T1): 1 patient experienced increasing shoulder complaints that were already present before the first visit. One patient was lost due to spontaneous severe back pain, which had also repeatedly occurred in the past. A third patient left the study due to personal circumstances. No other adverse events occurred during the study.

Instrumented gait assessments

Table III summarizes the group results for all measurements. There was a significant time effect on gait width (F(2,42)=6.143, p=0.005). Gait width increased by 12.6% from baseline to T1 (p=0.021), and this improvement persisted at T2 (9.7%; p=0.022) (see Fig. 2a). There was no effect of gait condition on gait width (F(1,21)=1.229, p=0.280), nor was there a significant time \times gait condition interaction (F(2,42)=1.081, p = 0.349).

Test	Before intervention (T0)	After 6 weeks (T1)	After 16 weeks (T2)	Friedman's test or repeated- measures ANOVA	<i>p</i> -value
GaitRite, mean (SD) [95% CI]					
Gait width – preferred speed, cm	10.7 (5.8) [8.1–13.2]	11.8 (6.0) [9.1–14.4]	11.3 (5.7) [8.7–13.8]	F(2, 42) = 6, 143	0 0059
Gait width – maximal speed, cm	10.0 (6.0) [7.4–12.7]	11.5 (6.1) [8.8–14.2]	11.2 (5.8) [8.7–13.8]	$\Gamma(2,42) = 0.143$	0.005
Comfortable gait speed, m/s	0.96 (0.25) [0.85-1.07)	1.04 (0.26) [0.93-1.16]	1.07 (0.28) [0.94-1.20]	F(2,42)=15.265	<0.001 ^b
Maximal gait speed, m/s	1.31 (0.41) [1.14-1.50]	1.33 (0.35) [1.17-1.48]	1.36 (0.41) [1.18-1.54]	F(2,42)=1.140	0.330
Radboud Falls Simulator					
Leg angle – known dir, mean (SD) [95% CI] 18.7 (4.1) [16.7–20.7]	19.8 (3.8) [17.9–21.6]	20.2 (4.1) [18.2-22,1]	F(2,32)=8.568	0.001 ^a
Leg angle – unknown dir, mean (SD) [95% CI] 19.1 (4.7) [16.3–22.0]	19.3 (4.7) [16.5–22.2]	19.4 (5.2) [16.3-22.6]	F(2,24)=0.107	0.899
Success rates – known dir, %, median [IQR] 70.0 [45.0]	90.0 [30.0]	90.0 [45.0]	$\chi^2(2, n=21)=12.559$	0.002 ^c
Success rates – unknown dir, %, median [IQR] 25.0 [55.0]	35.0 [72.5]	45.0 [70.0]	$\chi^2(2, n=22)=4.388$	0.111
Clinical assessments					
MAS – hip adductors, median [IQR]	2.5 [1.3]	1.0 [1.5]	1.8 [1.0]	$\chi^2(2, n=22)=33.890$	<0.001 ^d
MRC – hip adductors, median [IQR]	5.0 [0.0]	4.0 [1.0]	5.0 [0.3]	$\chi^2(2, n=22) = 15.800$	<0.001 ^e
MRC – hip abductors, median [IQR]	5.0 [0.1]	5.0 [1.0]	5.0 [0.0]	$\chi^2(2, n=22)=4.957$	0.084
ROM – hip abduction, mean (SD) [95% CI]	38.8 (9.4) [34.6-42.9]	50.7 (12.5) [45.2-56.2]	45.0 (10.8) [40.2-49.8]	F(2,42)=31.613	<0.001 ^f
Other functional tests					
BBS, median [IQR]	49.5 [10.0]	51.5 [9.0]	51.0 [7.8]	$\chi^2(2, n=22)=4.031$	0.133
TUG, s, mean (SD) [95% CI]	10.6 (3.8) [8.6–12.6]	10.7 (4.2) [8.6–12.9]	10.5 (4.3) [8.3–12.7]	F(2,42)=0.198	0.821
6MWT, s, mean (SD) [95% CI]	367.5 (87.9) [322.3-412.7]	376.0 (95.8) [327.0-425.2]	382.1 (95.2)[333.2-431.0]	F(2,32) = 2.498	0.098
ABC scale, %, mean (SD) [95% CI]	55.0 (21.85) [43.8-66.2]	54.7 (22.8) [43.0-66.4]	51.5 (23.7) [39.3-63.6]	F(2,42)=2.048	0.142

^aT0<T1/T2, T1=T2; ^bT0<T1<T2; ^cT0<T1, T0=T2, T1=T2; ^dT0>T2>T1; ^eT0/2>T1, T0=T2; ^fT0<T2<T1. MAS: Modified Ashworth Scale; MRC: Medical Research Council scale; ROM: range of motion; BBS: Berg Balance Scale; TUG: Timed Up and Go test; 6MWT: 6-min Walk Test; ABC: Activities-specific Balance Confidence scale; dir: direction.

Journal of Rehabilitation Medicine



Fig. 2. (a) Gait width and (b) gait speed at 3 measurements (T0, T1, T2) for preferred and maximal gait speed (gait condition). *Indicates significant *post-hoc* effects of time for gait width. #Indicates significant *post-hoc* effects of time on preferred gait speed.

There was also a significant *time* effect on gait speed (F(2,42)=5.458, p=0.008) as well as a significant *time* × gait condition interaction (F(2,42)=5.399, p=0.008). Compared with baseline, the preferred gait speed had increased by 8.3% at T1 (p=0.001) and by 11.5% at T2 (p<0.001). In contrast, no significant changes from baseline were observed in the maximal gait speed (Fig. 2b).

Instrumented balance assessments

Changes in leg angles across time could only be statistically tested if side steps were made at all measurements (T0–T2). As some participants failed to make any side steps (i.e. only made cross steps) or grabbed the railing system before stepping, a total of 19 participants could be included in the analysis of trials with known perturbation direction and 13 in the analysis of trials with unknown directions. The mean limit of stability was 2.135 m/s² (range 0.375–4.375 m/s²).

For the known direction perturbations, the leg angle showed a main effect of *time* (F(2,36) = 12.053,

p < 0.001). There was an increase of 5.9% in leg angle from baseline to T1 (p=0.003), which persisted at T2 (8.0%; p=0.001). In contrast, leg angle for the unknown direction perturbations did not show a significant effect of *time* (F(2,24)=0.107, p=0.899) (see Fig. 3a).

For the known direction perturbations, there was a significant effect of *time* on success rate ($\chi^2(2)=12.559$, p=0.002). Compared with the 70% success rate at baseline, participants were more successful at T1 (90%; p=0.007), which result tended to persist at T2 (90%; p=0.075). For the unknown direction perturbations, no significant effects of *time* were found ($\chi^2(2)=4.388$, p=0.111) (see Fig. 3b).

Physical tests

Hip adductor muscle tone showed a significant *time* effect ($\chi^2(2)=33.890$, p<0.001). The MAS scores decreased from baseline to T1 (p<0.001), and subsequently increased from T1 to T2 (p=0.001), although they did not reach the baseline values (T0 vs T2, p=0.001). Hip adductor muscle strength also showed a



Fig. 3. (a) Leg angle and (b) success rate at 3 measurements (T0, T1, T2) for known and unknown perturbation directions. #Indicates significant *post-hoc* effects of time in known perturbation directions. *Indicates significant effect of time.

JRV

significant effect of *time* (χ^2)=15.800, p<0.001), as the MRC scores decreased from baseline to T1 (p=0.005), and increased from T1 to T2 (p=0.003) to baseline values (T0 vs T2, p=0.480). Hip abductor muscle strength did not significantly differ between the 3 measurements ($\chi^2(2)$ =4.957, p=0.084). Furthermore, there was a significant *time* effect on hip abduction ROM (F(2,42)=31.613, p<0.001), which increased from baseline to T1 (p<0.001) and decreased from T1 to T2 (p=0.001), although it was still above baseline values at T2 (T0 vs T2, p<0.001).

Functional tests

Because the 6MWT was too demanding for 5 participants, the analysis was performed on the remaining 17 participants. This analysis showed no significant effect of *time* on the 6MWT (F(2,32)=2.498, p=0.098). Likewise, the BBS ($\chi^2(2)=4.031, p=0.133$), TUG (F(2,42)=0.198, p=0.821), and ABC scale (F(2,42)=2.048, p=0.142) also did not show significant *time* effects.

DISCUSSION

The aim of this study was to evaluate the effects of BTX-A treatment and subsequent stretching of the hip adductors on gait and balance capacities in patients with pure HSP. The results support our hypothesis that bilateral BTX-A injections in, and subsequent stretching of, the hip adductors improve both gait width and lateral stepping responses in known perturbation directions coinciding with reduced hip adductor tone 6 weeks after treatment compared with baseline. These functional effects were retained, although some recurrence of hip adductor muscle tone was found 10 weeks later. In addition, comfortable gait speed increased 6 weeks post-injections, which effect was even a little stronger after 16 weeks. In contrast, maximal gait speed and lateral stepping responses in unknown perturbation directions did not respond to the treatment, nor did other functional tests (BBS, TUG, 6MWT, ABC scale).

Besides a prolonged reduction in hip adductor muscle tone lasting up to 16 weeks after treatment, we observed a similar and substantial improvement in hip abduction ROM (on average 13.3°) and a temporary reduction of hip muscle strength (on average 1 point on the MRC scale, 6 weeks after treatment). These findings are in line with several previous studies (2, 3, 5, 6, 8, 9, 11). The mean improvement of muscle tone at 6 weeks post-injections was 1.5 point on the MAS, which can be considered substantial and clinically relevant (22). Interestingly, the observed temporary loss of muscle strength did not seem to have a detrimental effect on balance and gait capacities, since functional tasks improved (or remained stable) 6 weeks after treatment. This result is in agreement with a previous study from our group (2) and suggests that loss of muscle strength after BTX-A injections in spastic muscles is probably of relatively short duration and without noticeable functional disadvantage.

The observed improvement in gait width was, on average, 1.1 and 1.5 cm for comfortable and maximal walking speed, respectively, and was probably the direct result of reduced hip adductor tone and improved hip abduction ROM. Remarkably, this effect was found even though comfortable gait speed improved in parallel. It is conceivable that a higher gait speed might coincide with faster leg swing and, thus, aggravate velocity-dependent hip adductor spasticity, but this effect apparently did not occur. Instead, the reason why comfortable gait speed improved may be that reduced hip adductor tone allowed patients to make larger and/or faster steps. In a prior study of patients with spastic paraparesis of various origins, BTX-A injections in the hip adductors also led to increased gait velocity (23). In addition, previous studies of patients with HSP in which hip adductors were injected with BTX-A in combination with other muscle groups also showed increased gait speed (6, 8, 9). These findings have important clinical implications as reduced spontaneous gait speed is one of the most frequent problems reported by patients with HSP (24). Notably, maximal gait speed did not improve after treatment, which is in agreement with the notion that the hip adductors are not the key muscles for gait propulsion (25). Nevertheless, gait width during walking at maximal gait speed improved even more strongly than during comfortable walking, indicating the robustness of the treatment effects on widening the base of support while walking. The observed effect size for gait width in this study may seem small, but it constitutes a 10-15% increase compared with baseline. Hence, the observed increase in gait width of 1.1–1.5 cm may be clinically relevant in terms of improved frontal-plane balance, but this conclusion needs to be supported by future studies.

To our knowledge, the effects of bilateral BTX-A injections on reactive lateral stepping responses have not been investigated previously. In the present study, we only observed beneficial effects on stepping responses to perturbations in known directions. Interestingly, no improvements in stepping leg angle or success rate were observed upon perturbations in unknown directions. As hip adductor spasticity decreases, less muscle activity of the hip abductors is supposedly needed to make a lateral step. One explanation for the observed discrepancy between known and unknown perturbation direction may be that the individual maximal perturbation intensity at

440 B. J. H. van Lith et al.

baseline was based on tests with a known perturbation direction. As a consequence, the perturbation intensities might have been too high for the participants following perturbations in unknown directions. This may also (partly) explain the much lower number of participants able to sustain perturbations with unknown compared with known directions. Another reason might be that reactive steps following unexpected perturbations are relatively strongly influenced by delayed postural responses that occur in patients with HSP(4, 26), because patients cannot compensate by an anticipatory "central set" if the direction is unknown. Nevertheless, the improved balance capacity following perturbations in known directions suggests that hip adductor spasticity indeed impairs the quality of lateral stepping responses and, thus, is a relevant treatment target from a frontalplane balance perspective.

The fact that the beneficial effects on muscle tone, gait width, comfortable gait velocity, lateral balance persisted until 16 weeks after treatment seems to challenge the common opinion that the biological effects of BTX-A have worn off after this time interval. It may, therefore, be that the stretching component of the treatment protocol was responsible for the observed long-term effects.

Study limitations and future perspectives

A limitation of the present exploratory study in patients with HSP is the relatively small sample size and the lack of a control condition, while the rather stringent inclusion and exclusion criteria limit the generalizability of our findings. Nevertheless, this study provides indications for the beneficial effects of bilateral BTX-A injections in the adductor longus, adductor magnus and gracilis muscles and subsequent stretching of these muscles on gait width, comfortable gait speed, and reactive lateral stepping in known perturbation directions, whereas maximal gait speed, gait endurance, and clinical balance scores appear to be less responsive in these patients. Future research should preferably be multi-centred to increase the number of participants and use a randomized controlled design. Gait width and gait speed would be valuable and responsive outcome measures in such trials. The instrumented balance assessments used in the present study are, however, less suitable for multi-centre studies as they require further development of clinically affordable systems. Clinically applicable assessments to validly test the quality of (lateral) reactive stepping responses are therefore urgently needed. As HSP is a chronic and progressive condition, it would also be relevant to conduct longitudinal, comparative cohort studies to investigate whether repetitive cycles of BTX-

A treatment of spastic hip adductors improve lifetime functional ambulation prognosis in these patients.

ACKNOWLEDGEMENT

This investigation was supported by an unrestricted grant from Merz Pharmaceuticals.

The authors have no conflicts of interest to declare.

REFERENCES

- Hesse S, Werner C, Bardeleben A, Brandl-Hesse B. Management of upper and lower limb spasticity in neurorehabilitation. Acta Neurochir Suppl 2002; 79: 117–122.
- de Niet M, de Bot ST, van de Warrenburg BP, Weerdesteyn V, Geurts AC. Functional effects of botulinum toxin type-A treatment and subsequent stretching of spastic calf muscles: a study in patients with hereditary spastic paraplegia. J Rehabil Med 2015; 47: 147–153.
- Gracies JM, Singer BJ, Dunne JW. The role of botulinum toxin injections in the management of muscle overactivity of the lower limb. Disabil Rehabil 2007; 29: 1789–1805.
- Nonnekes J, van Lith B, van de Warrenburg BP, Weerdesteyn V, Geurts ACH. Pathophysiology, diagnostic work-up and management of balance impairments and falls in patients with hereditary spastic paraplegia. J Rehabil Med 2017; 49: 369–377.
- Riccardo M, Angela L, Angela D, Vita P, Giulio L, Pietroq F, et al. Combined treatment Fkt-botulinum toxin type A (Btx-A) in patients with Strumpell-Lorrain disease. Curr Pharm Des 2016; 22: 758–763.
- Rousseaux M, Launay MJ, Kozlowski O, Daveluy W. Botulinum toxin injection in patients with hereditary spastic paraparesis. Eur J Neurol 2007; 14: 206–212.
- Servelhere KR, Faber I, Saute JA, Moscovich M, D'Abreu A, Jardim LB, et al. Non-motor symptoms in patients with hereditary spastic paraplegia caused by SPG4 mutations. Eur J Neurol 2016; 23: 408–411.
- Hecht MJ, Stolze H, Auf dem Brinke M, Giess R, Treig T, Winterholler M, et al. Botulinum neurotoxin type A injections reduce spasticity in mild to moderate hereditary spastic paraplegia--report of 19 cases. Mov Disord 2008; 23: 228–233.
- Dunne JW, Heye N, Dunne SL. Treatment of chronic limb spasticity with botulinum toxin A. J Neurol Neurosurg Psychiatry 1995; 58: 232–235.
- Wissel J, Ward AB, Erztgaard P, Bensmail D, Hecht MJ, Lejeune TM, et al. European consensus table on the use of botulinum toxin type A in adult spasticity. J Rehabil Med 2009; 41: 13–25.
- Kaji R, Osako Y, Suyama K, Maeda T, Uechi Y, Iwasaki M, et al. Botulinum toxin type A in post-stroke lower limb spasticity: a multicenter, double-blind, placebo-controlled trial. J Neurol 2010; 257: 1330–1337.
- Cutlip RG, Mancinelli C, Huber F, DiPasquale J. Evaluation of an instrumented walkway for measurement of the kinematic parameters of gait. Gait Posture 2000; 12: 134–138.
- Nonnekes J, de Kam D, Geurts AC, Weerdesteyn V, Bloem BR. Unraveling the mechanisms underlying postural instability in Parkinson's disease using dynamic posturography. Expert Rev Neurother 2013; 13: 1303–1308.
- Davis RB, Ounpuu S, Tyburski D, Gage JR. A gait analysis data-collection and reduction technique. Hum Mov Sci 1991; 10: 575–587.
- 15. Weerdesteyn V, Laing AC, Robinovitch SN. The body configuration at step contact critically determines the successfulness of balance recovery in response to large backward perturbations. Gait Posture 2012; 35: 462–466.
- 16. Council MR. Aids to the examination of the peripheral

nervous system, London: Her Majesty's Stationary Office; 1981.

- Bohannon RW, Smith MB. Interrater reliability of a modified Ashworth scale of muscle spasticity. Phys Ther 1987; 67: 206–207.
- Berg KO, Wood-Dauphinee SL, Williams JI, Maki B. Measuring balance in the elderly: validation of an instrument. Can J Public Health 1992; 83 Suppl 2: S7–S11.
- Podsiadlo D, Richardson S. The timed "Up & Go": a test of basic functional mobility for frail elderly persons. J Am Geriatr Soc 1991; 39: 142–148.
- Butland RJ, Pang J, Gross ER, Woodcock AA, Geddes DM. Two-, six-, and 12-minute walking tests in respiratory disease. Br Med J (Clin Res Ed) 1982; 284: 1607–1608.
- Powell LE, Myers AM. The Activities-specific Balance Confidence (ABC) Scale. J Gerontol A Biol Sci Med Sci 1995; 50A: M28–M34.
- 22. Esquenazi A, Albanese A, Chancellor MB, Elovic E, Segal KR, Simpson DM, et al. Evidence-based review and as-

sessment of botulinum neurotoxin for the treatment of adult spasticity in the upper motor neuron syndrome. Toxicon 2013; 67: 115–128.

- Takenaga S, Kawahigashi Y, Sonoda Y, Horikiri T, Hirata K, Arimura K, et al. [Treatment of spastic paraparesis with botulinum toxin with reference to beneficial effects, disease severity and long-term treatment.] Rinsho Shinkeigaku 1995; 35: 251–255 (in Japanese).
- 24. Klebe S, Stolze H, Kopper F, Lorenz D, Wenzelburger R, Volkmann J, et al. Gait analysis of sporadic and hereditary spastic paraplegia. J Neurol 2004; 251: 571–578.
- Ellis RG, Sumner BJ, Kram R. Muscle contributions to propulsion and braking during walking and running: insight from external force perturbations. Gait Posture 2014; 40: 594–599.
- Nonnekes J, de Niet M, Oude Nijhuis LB, de Bot ST, van de Warrenburg BP, Bloem BR, et al. Mechanisms of postural instability in hereditary spastic paraplegia. J Neurol 2013; 260: 2387–2395.

Journal of Rehabilitation Medicine